Author's response to reviews

Title: Use and acceptance of long lasting insecticidal net screens for dengue prevention in Acapulco, Guerrero, Mexico

Authors:

Catrin H Jones (catrinhuwsjones@gmail.com)
David Benitez-Valladares (davidbeniva@gmail.com)
Guillermo Guillermo-May (ggm.biol@gmail.com)
Felipe Dzul-Manzanilla (fdzul@me.com)
Azael Che-Mendoza (achemendoza_vecrokes@hotmail.com)
Mario Barrera-Pérez (barreraperezma@gmail.com)
Johannes Sommerfeld (sommerfeldj@who.int)
Axel Kroeger (kroegera@who.int)
Timothy O'Dempsey (T.Odempsey@liverpool.ac.uk)
Pablo Manrique-Saide (pablo_manrique2000@hotmail.com)
Celia Selem-Salas (ssalas@uady.mx)
Juan Chablé-Santos (jcsantos@uady.mx)

Version: 2
Date: 18 March 2014

Author's response to reviews: see over
The changes that have been made are the following:

1.) The email address of all the authors on the Title page have been provided.

2.) **Consent Statement:** Written informed consent for participation in the study was obtained from participants.

3.) **Ethics Committee approval:** The LSTM Masters Ethics Committee approved this study, and the Internal Review Board of UADY and the Ethical Review Committee of the WHO approved the larger UADY study, including this work.